Author's response to reviews

Title: Development and psychometric properties of a new social support scale for self-care in middle-aged patients with type II diabetes (S4-MAD)

Authors:

Shohreh Naderimagham (naderimagham@modares.ac.ir)
Shamseddin Niknami (niknamis@modares.ac.ir)
Farid Abolhassani (abolhassanif@tums.ac.ir)
Ebrahim Hajizadeh (hajizadeh@modares.ac.ir)
Ali Montazeri (montazeri@acecr.ac.ir)

Version: 6 Date: 20 October 2012

Author's response to reviews: see over
Dear Victorino,

MS: 1080646941697410
Development and psychometric properties of a new social support scale for self-care in middle-aged patients with type II diabetes (S4-MAD)
BMC Public Health

Thank you for your e-mail. As you see in the following reports there are nothing to respond. In fact it seems that both reviewers were satisfied with our previous revisions.

Reviewer's report
Title: Development and psychometric properties of a new social support scale for self-care in middle-aged patients with type II diabetes (S4-MAD)
Version: 5 Date: 25 September 2012
Reviewer: Sepideh Omidvari
Reviewer's report:
Major Compulsory Revisions: ___
Minor Essential Revisions: ___
Discretionary Revisions: ___
Level of interest: An article whose findings are important to those with closely related research interests
Quality of written English: Acceptable
Statistical review: No, the manuscript does not need to be seen by a statistician.
Declaration of competing interests:
I declare that I have no competing interests

Reviewer's report
Title: Development and psychometric properties of a new social support scale for self-care in middle-aged patients with type II diabetes (S4-MAD)
Version: 5 Date: 27 September 2012
Reviewer: Min-Feng Huang
Reviewer's report:
Minor Essential Revisions
Level of interest: An article of limited interest
Quality of written English: Acceptable
Statistical review: No, the manuscript does not need to be seen by a statistician

I wish you all the best.
Kind regards
Ali Montazeri